Author's response to reviews

Title: Antinociceptive and anti-inflammatory activities of Geranium bellum and its isolated compounds

Authors:

Claudia Gonzalez (cvg09@yahoo.com)
Raquel Cariño (raquelcarchortes@gmail.com)
Juan A Gayosso (gayitojuan@yahoo.com)
Mario I Ortiz (mario_i ortiz@hotmail.com)
Minarda De la O (minidelao@hotmail.com)
Diana A Altamirano (baezaltaandrea@hotmail.com)
Luis Jimenez (luis.jimenez@ieee.org)
Mirandeli Bautista (mirandeli@hotmail.com)

Version: 4
Date: 24 October 2014

Author's response to reviews: see over
In response to your e-mail of date 14/10/2014 we are submitting a modified version of our paper:

MS: 1726423595103322

Title: Antinociceptive and anti-inflammatory activities of *Geranium bellum* and its isolated compounds.

Authors: Claudia Velázquez, Raquel Cariño, Juan A Gayoso, Mario I Ortiz, Minarda De la O, Diana A Altamirano, Luis Jimenez and Mirandeli Bautista

Reviewer: Lucia Helena Faccioli

Reviewer’s report:

Almost of questions raised were done. However, there are still minor errors in the text that need corrections:

1. Please standardize *Geranium bellum* for *G. bellum* after the first show. Please see subtitle in the Figures 4 and 6.

   The abbreviated genus is now used after the first full mention of the species.

2. In the Figure 4: change percentage inhibition from 38.35% for 38%; 42.46% for 43%, etc.

   This point was corrected.

3. There are errors in author names in the references. Examples: see references 12 and 14: chance Rogerio PA for Rogerio AP. Do the same for the others authors. Please check all names in all the references.

   We have addressed this suggestion.

4. There have concordance on the number of animals used. In Methods is written 6-8 animals, but there are not 8 rats or mice in the figures.

   We have added the exact “n” value.